Author’s response to reviews

Title: The impact of a Referral Card-Based Intervention on Intimate Partner Violence, Psychosocial Health, Help-Seeking and Safety Behaviour during Pregnancy and Postpartum: A Randomized Controlled Trial

Authors:

An-Sofie Van Parys (An-Sofie.VanParys@uzgent.be; ansofie.vanparys@ugent.be)
Ellen Deschepper (ellen.deschepper@ugent.be)
Kristien Roelens (kristien.roelens@ugent.be)
Marleen Temmerman (marleen.temmerman@ugent.be)
Hans Verstraelen (hans.verstraelen@ugent.be)

Version: 4 Date: 28 Aug 2017

Author’s response to reviews:

Dear editor,

In response to your email d.d. August 22th, 2017, we revised the manuscript and incorporated the suggestions made.

Underneath, we provide an overview of all adjustments made:

1. We have addressed your concern of overlap once more and rephrased the marked sections (where possible) and have added a summarizing statement in the methodology section;

2. We have changed the “Conclusion and recommendations” section to “Conclusions”

3. We have changed the “Objectives” section to “Background” in the Abstract.

4. We have also uploaded a single, final, clean version of the manuscript.

We hope these adaptations will facilitate the swift publication of our manuscript.

Thank you for your time and consideration.
Yours sincerely,

Dr. An-Sofie Van Parys,
Ghent University, Belgium